ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement ("Agreement") is made and entered into this _______ day of July, 2023, by and between Indiana-American Water Company, Inc., an Indiana public utility corporation ("Buyer"), and Silver Creek Water Corporation, an Indiana non-profit utility corporation ("Seller"). Hereinafter, the Buyer and Seller may be individually referred to as a "party" or jointly as the "parties".

RECITALS:

- A. Seller owns and operates a water system which provides potable water service to customers located within its service area, graphically depicted on <u>Exhibit A</u> attached hereto, and incorporated herein (the "Service Area") (collectively referred to as the "Business").
- B. Buyer desires to acquire and Seller desires to sell the Assets of Seller relating to the Business pursuant to the terms and conditions of this Agreement.

NOW, THEREFORE, in consideration of the foregoing recitals and the covenants contained herein and in exchange for other consideration the receipt and sufficiency of which are hereby acknowledged, the parties, intending to be legally bound, agree as follows:

ARTICLE 1 Definitions and Related Matters

For purposes of this Agreement and all documents executed in connection with this Agreement, the capitalized terms shall have the meanings assigned to them herein or in Exhibit B.. All references to "Schedules" in this Agreement mean the separately numbered schedules corresponding to the section numbers in this Agreement that are contained in a set of disclosure schedules (the "Disclosure Schedules") delivered contemporaneously with the execution of this Agreement, which will contain information supplemental to this Agreement and information constituting factual exceptions to the representations and warranties of the Seller contained herein. Capitalized terms used in the Disclosure Schedules and not otherwise defined therein will have the meanings given to them in this Agreement, including Exhibit B hereto.

ARTICLE 2 Purchase and Sale of Assets; Closing

- 2.1 Transfer and Description of Assets. Subject to and upon all other terms and conditions of this Agreement, effective as of the Effective Time on the Closing Date, Seller shall sell, convey, transfer, assign and deliver to Buyer free and clear of all Encumbrances, and Buyer shall acquire from Seller, all of Seller's right, title and interest in and to all of Seller's assets, properties, business, and right of every kind and description, whether tangible or intangible, real, personal or mixed, specifically including but not limited to all assets listed in the Public Water System Facilities Inventory/Valuation of the Silver Creek Water Corporation (SCWC) Water System, dated November 2022 (the "Appraisal"), other than the Excluded Assets, regardless of where located, which are, should be, could be, or in the future would be part of the water system utilized to provide potable water service to Seller's customers, including the following:
- (a) all Real Property, including the office building, garage and storage buildings and any easements, rights-of-way, licenses, or other rights or interests in property granted to, held by, or owned by the Seller in furtherance of the operation of the Business, including, but not limited to those set forth on Schedule 3.4(c);
- (b) all Tangible Personal Property, as defined in <u>Exhibit B</u>, utilized by Seller in the operation of the Business;
- (c) all Data and Records related to Seller's operation of the Business, including the customer list which shall include the service and billing address of all customers of Seller and, subject to applicable Law, copies of all Records related to Seller's operation of the Business, except for personnel records and as described in Section 2.2(b);
 - (d) the Material Contracts set forth on <u>Schedule 2.1(d)</u> (the "Assigned Contracts");

- (e) all Permits and all pending applications therefor, renewals thereof or exemptions therefrom which are necessary or advisable in the operation of Seller's Business, including those listed in <u>Schedule 3.9</u>; and
- (f) all of the intangible rights and property, and/or interests in property, of Seller utilized by Seller in the operation of the Business.

All of the foregoing shall be hereinafter referred to collectively as the "Assets".

- **2.2** Excluded Assets. Notwithstanding anything to the contrary contained in Section 2.1 or elsewhere in this Agreement, the following assets of Seller are not part of the sale and purchase contemplated hereunder, are excluded from the Assets, and shall remain the property of Seller after the Closing:
 - (a) all insurance policies and rights thereunder;
- (b) all personnel Records and other Records that Seller is required by Law to retain in its possession;
- (c) all rights in connection with and assets of Seller's Plans, Seller's Benefit Obligations and employment or independent contractor Contracts;
 - (d) all Contracts other than the Assigned Contracts;
 - (e) all rights of Seller under the Transaction Documents;
 - (f) cash, cash equivalents, short-term investments, and customer deposits;
 - (g) accounts receivable arising prior to the Effective Time; and
- (h) Customer Service Connections, which shall remain the property of the customer.

All of the foregoing shall be hereinafter referred to collectively as the "Excluded Assets".

- **2.3 Consideration.** The consideration for the Assets will be \$45,000,000. The Assets shall be in substantially the same condition at the time of Closing, as represented by Seller in Article 3. If any of the Assets materially deviate from the condition represented by Seller in Article 3 and are not able to perform the function they are intended to, the Buyer, at its option, may request an adjustment to the Purchase Price to compensate for the deterioration or loss of said Asset, but Seller is under no obligation to agree to any requested adjustment.
- **2.4 Liabilities.** The Buyer shall not be responsible for any of the Liabilities of Seller, including any that may arise after Closing, and any such Liabilities shall remain the sole responsibility of and shall be retained, paid, performed and discharged solely by Seller except as provided in the following sentence. Subject to the terms and conditions set forth herein, Buyer shall assume and agree to pay, perform and discharge when due any and all liabilities and obligations arising out of or relating to Buyer's ownership of the Assets and operation of the Business after the Closing.
- **2.5** Closing. The purchase and sale provided for in this Agreement will take place at a location and time of day agreed upon by the parties (the "Closing"). The date of the Closing shall be after the satisfaction of all conditions precedent to the closing contained in this Agreement, including the Indiana Utility Regulatory Commission (the "IURC") issuing an Order approving the Contemplated Transaction and the terms and conditions contained in Section 5.1 (d) herein. Closing shall be effective as of 5:00 pm local time (the "Effective Time") on the actual date of Closing (the "Closing Date").

2.6 Closing Obligations.

- (a) At or prior to Closing, Seller shall deliver to Buyer the following documents, duly executed:
- (i) a Bill of Sale for all of the Assets that are Tangible Personal Property in a form reasonably acceptable to Buyer ("Bill of Sale");

- (ii) an assignment of all of the Assets that are intangible personal property in a form reasonably acceptable to Buyer;
- (iii) for each interest in Real Property identified on <u>Schedule 3.4(c)</u>, or discussed in Section 2.1 above, a recordable warranty deed or such other appropriate document or instrument of transfer or approval, as the case may require, each in form and substance reasonably satisfactory to Buyer and its legal counsel. Notwithstanding the fact that the same may not be listed on <u>Schedule 3.4(c)</u>, Seller must provide easements or other transferable property rights, licenses, or interests in property to Buyer for all mains used in the Business, which are not located on or in public rights-of-way, and must provide assignments of public rights-of-way Permits with only those conditions acceptable to Buyer for all mains located in municipal, county or state owned public rights-of-way;
- (iv) such other deeds, bills of sale, assignments, certificates of title, documents and other instruments of transfer and conveyance as may be reasonably requested by Buyer, each in form and substance satisfactory to Buyer;
- (v) a certificate as to the accuracy of Seller's representations and warranties as of the date of this Agreement and as of Closing in accordance with Section 5.1(a) and as to its compliance with and performance of its covenants and obligations to be performed or complied with at or before Closing in accordance with Section 5.1(b);
- (vi) a certificate of the Seller's President, dated as of the Closing, certifying: (A) that attached are true copies of the duly adopted resolutions of the board of directors of Seller authorizing the execution of this Agreement and the sale of all Assets; and (B) the incumbency, signatures and authority of the officer or officers of Seller executing this Agreement or any agreement contemplated hereby on behalf of Seller;
- (vii) an assignment and assumption agreement in a form reasonably acceptable to Buyer (the "Assignment and Assumption Agreement") effecting the assignment to and assumption by Buyer of the Assets; and
- (b) At or prior to Closing, Buyer shall deliver to Seller, the following documents, duly executed, or funds:
- (i) The Purchase Price, by wire transfer or other immediately available funds to an account specified by the Seller;
- (ii) a certificate as to the accuracy of Buyer's representations and warranties as of the date of this Agreement and as of Closing in accordance with Section 5.2(a) and as to its compliance with and performance of its covenants and obligations to be performed or complied with at or before Closing in accordance with Section 5.2(b);
- (iii) a certificate of the company secretary of Buyer, dated as of the Closing, certifying: (A) that attached are true copies of the duly adopted resolutions of the Buyer's board of directors authorizing the execution of this Agreement and the purchase of all Assets; and (B) the incumbency, signatures and authority of the officer or officers of Buyer executing this Agreement or any agreement contemplated hereby on behalf of Buyer;
- (iv) a certificate of existence issued by the Secretary of State of Indiana with respect to Buyer, dated not earlier than thirty (30) days prior to Closing; and
- **2.7** Applicable Rates. On and after the Closing, the customers of Seller shall be subject to all of Buyer's rates and charges generally applicable in Area One on Seller's schedule of rates and charges, as the same may be approved by the IURC from time to time. Applicable charges include DSIC, SEI and any other charge approved by the Commission.

ARTICLE 3 Representations and Warranties of Seller

Seller hereby makes the following representations and warranties to Buyer as of the date of this Agreement and as of the Closing:

3.1 Organization. Seller is a nonprofit utility corporation duly organized, and validly existing under the Laws of the State of Indiana. Seller has the power and authority to own, lease, sell and operate its assets (including the Assets) and to conduct the Business as it is now being conducted.

3.2 Enforcement: Authority: No Conflict.

- (a) Seller has the power to enter into and perform this Agreement and the other Transaction Documents to which it is a party and to consummate the Contemplated Transaction.
- Transaction Documents to which it is a party by Seller, and the consummation of the Contemplated Transaction, have been duly and validly authorized by all necessary action on the part of Seller except that, as of the date of this Agreement, Seller has not yet obtained the approval of its members to the performance of this Agreement and the consummation of the Contemplated Transaction. This Agreement has been duly executed and delivered by Seller (and all documents required to be executed and delivered by Seller at Closing shall be duly executed and delivered by Seller), and this Agreement constitutes (subject to the member approval exception described in the preceding sentence), and at the Closing this Agreement and such documents shall constitute, the valid and binding obligations of Seller, enforceable in accordance with their terms except as such enforceability may be limited by (i) bankruptcy, insolvency, reorganization, moratorium or similar Laws now or hereafter in effect relating to creditors' rights generally and (ii) equitable defenses at the discretion of the court before which any Proceeding therefor may be brought (clauses (i) and (ii) collectively, the "Bankruptcy and Equity Exceptions").
- (c) Neither the execution, delivery and performance of this Agreement and the other Transaction Documents to which Seller is a party by Seller, nor the consummation of the Contemplated Transaction will, directly or indirectly (with or without notice or lapse of time):
- (i) contravene, conflict with or result in a violation of: (A) any provision of the Organizational Documents of Seller; or (B) any resolution adopted by the governing body of Seller;
- (ii) contravene, conflict with or result in a violation of any Laws or any Order to which Seller or any of the Assets may be subject, or require the consent of or other action by a Governmental Authority except for the IURC and the Indiana Department of Environmental Management (the "IDEM");
- (iii) contravene, conflict with or result in a violation of any of the terms or requirements of or give any Governmental Authority the right to revoke, withdraw, suspend, cancel, terminate or modify any Permit or other authorization by a Governmental Authority that is held by Seller or that otherwise relates to the Business or any of the Assets;
- (iv) except for the requirement to obtain consent of a third party to transfer the Assigned Contracts described in Schedule 3.2(c)(iv), contravene, conflict with or result in a default under, or give rise to any right of termination, cancellation or acceleration of any right or obligation of Seller or to a loss of any benefit to which Seller is entitled under any provision of any Assigned Contract to which Seller is a party, or by which any of the Assets or any other material assets and properties of Seller are bound or subject; or
- (v) result in the imposition or creation of any Encumbrance upon or with respect to any of the Assets, except for Permitted Encumbrances or as contemplated by this Agreement.

3.3 Assets.

- (a) Seller has good and marketable title to, valid leasehold interest in or valid licenses to use, all of the Assets, free and clear of all Encumbrances, other than Permitted Encumbrances. The use of the Assets is not subject to any Encumbrances, other than Permitted Encumbrances, and such use does not encroach on the property or the rights of any Person.
- (b) The Assets are sufficient for, and constitute all the assets, properties, business, and rights of every kind and description, and services required for, the continued conduct and operation of the Business by Buyer in substantially the same manner as currently conducted and

operated by Seller. The Assets, taken as a whole, comprise all the assets, properties, business, and rights of every kind and description used or held for use in, or useful or necessary to the operation of the Business as currently operated by Seller. There are no Assets that are owned by any Person other than Seller that will not be licensed or leased to Buyer at Closing under valid license arrangements or leases on terms no less favorable to Buyer than the terms applicable to Seller as of the date hereof under such license agreements or leases. None of the Excluded Assets are material to the Business.

(c) All Assets appraised in preparation for the Contemplated Transaction are in substantially the same condition as they were at the time of their valuation for the Appraisal, subject to Seller's repair or replacement of such Assets as needed in the ordinary course of Seller's business, consistent with past practice. Except as otherwise disclosed to Buyer in Seller's five-year plan, leak records, and other Records, all other Assets are in good operating condition and repair (except for normal wear and tear) and are usable in the ordinary course of the business and are being operated in conformity in all material respects with all applicable Laws and the terms of any Contracts to which Seller is a party or by which such Assets are subject or bound.

3.4 Real Property.

- (a) With respect to all real property owned by Seller in connection with its operation of the Business (including Seller's ownership rights in any easements and rights-of-way granted to Seller) (the "Owned Real Property"), there is no condemnation, expropriation or other Proceeding in eminent domain pending or, to the Knowledge of Seller, threatened or contemplated affecting any parcel of Owned Real Property or any interest therein. Seller has made available to Buyer all title insurance policies, property condition reports, and surveys for each parcel of Owned Real Property. There are no outstanding options, rights of first offer, or rights of first refusal to purchase any Owned Real Property or any portion thereof or interest therein. Seller is not a party to any agreement or option to purchase any real property or interest therein.
- (b) Seller has made available to Buyer a correct and complete copy of each lease, sublease, license or other Contract, currently in effect and as amended to date (each, a "Lease"), under which any real property leased or subleased has been granted to Seller in connection with its operation of the Business (including any easements and rights-of-way) (the "Leased Real Property" and together with the Owned Real Property, the "Real Property"). Neither Seller nor, to the Knowledge of Seller, any other party to any such Lease, is in breach or default, and no event has occurred (including the failure to obtain any consent) which, with notice or lapse of time or both, would constitute a breach or default under or permit termination or material modification of, or acceleration of a material amount of rents due under, any Lease.
- (c) All interests in Owned Real Property have been duly recorded. All easements included in Owned Real Property are enforceable pursuant to their terms against the subservient estate.
 - (d) The Real Property is set forth on <u>Schedule 3.4(d)</u>
- **3.5** <u>Financial Statements</u>. True, correct and complete copies of Seller's audited financial statements for the calendar year ended December 31, 2022 are available.

3.6 <u>Taxes</u>.

- (a) Seller has filed all returns for Taxes required to be filed under applicable Law arising from its ownership or operation of the Business, and all such Tax returns were correct and complete in all material respects.
- (b) Seller has timely paid all material Taxes due from or with respect to it arising from its ownership or operation of the Business, except Taxes that are being contested in good faith as set forth on <u>Schedule 3.6</u>.
- (c) Seller has withheld and paid all material Taxes required to have been withheld and paid in connection with amounts paid or owing any employee, independent contractor, creditor, stockholder or other third Person, and all Forms W-2 and 1099 required with respect thereto have been properly and timely filed or provided.

- (d) No Proceeding or audit is now in progress with respect to Seller regarding Taxes, and Seller has not received written notice of any pending or threatened Proceeding or audit against it (which remains outstanding) from any applicable Governmental Authority.
- (e) Seller has not waived any statute of limitations in respect of any material Taxes or agreed to any extension of time with respect to a material Tax assessment or deficiency.
- (f) Seller is not, nor has been, a party to any understanding or arrangement described in Section 6662(d)(2)(C)(ii) of the Code or any "listed transaction" as defined in Section 6707A(c)(2) of the Code and Treasury Regulation § 1.6011-4(b)(2).
- Contracts. Set forth on Schedule 3.7 is a complete and correct list of all Contracts (a) by which any of the Assets are bound or affected or (b) related to the Business to which Seller is a party (the "Material Contracts"). Seller has delivered or caused to be delivered to Buyer correct and complete copies of each Material Contract (including all amendments thereto) or a description of the terms of each Material Contract which is not in writing, and all documents affecting the rights or obligations of any party thereto. The Material Contracts have not been modified or amended except as disclosed on Schedule 3.7. Each Material Contract is valid and enforceable against Seller in accordance with its terms and is in full force and effect, and to Seller's Knowledge each Material Contract constitutes a legal, valid and binding obligation of the other parties thereto, enforceable against them in accordance with its terms. To Seller's Knowledge, no default and no event which, with the giving of notice, lapse of time, or both, would be a default has occurred under any Material Contract. To Seller's Knowledge, there are no setoffs, counterclaims or disputes existing or asserted with respect to such Material Contracts, and Seller has not made any agreement with any other party thereto for any deduction from or increase to any amount payable thereunder. To Seller's Knowledge, there are no facts, events or occurrences which in any way impair the validity or enforcement of any Material Contract or tend to reduce or increase the amounts payable thereunder. Seller has not, directly or indirectly, by operation of Law or otherwise, transferred or assigned all or any part of its right, title or interest in and to any Material Contract to any other Person. To Seller's Knowledge, there are no Proceedings pending nor threatened against any party to any of the Material Contracts

3.8 Environmental Matters.

Except as set forth on Schedule 3.8, to the Knowledge of Seller:

- (a) Seller has at all times complied with all applicable Environmental Laws, except where failure to comply would not, individually or in the aggregate, result in material Liabilities under Environmental Law.
- (b) Seller has not received any court order, demand, notice, or other written communication relating to any (i) actual, alleged, or potential violation of or failure to comply with any Environmental Law (ii) actual or potential Liability resulting from or arising under any Environmental Law or otherwise relating to Hazardous Materials; or (iii) actual or potential Liability with respect to any off-site Environmental Law matter including but not limited to, Hazardous Materials handling, transportation, treatment, storage, management or disposal services facilities or locations used by Seller.
- (c) There are no pending or threatened Liabilities under any Environmental Law, with respect to, arising from, or affecting any of the Real Property or any other asset owned or used by Seller.
- (d) There is no Hazardous Material present on or under any Real Property owned by Seller in material violation of Environmental Law, and there is no Hazardous Material present on or under any of the other Real Property in material violation of Environmental Law.
- (e) None of the Real Property owned by Seller contains any (i) above-ground or underground storage tanks or (ii) landfills, surface impoundments, or Hazardous Materials disposal areas. None of the other Real Property contains any (i) above-ground or underground storage tanks or (ii) landfills, surface impoundments, or Hazardous Materials disposal areas.
- (f) No Hazardous Materials have been transported, discharged, released, spilled or disposed from the Real Property owned by Seller (or, from any of the other Real Property) during Seller's occupancy of the Real Property in material violation of any Environmental Law; no Hazardous

Materials have been generated, handled, treated, stored, managed or disposed of at, on or under the Real Property owned by Seller (or, from any of the other Real Property) during Seller's occupancy of such Real Property in material violation of any Environmental Law; and there has been no release of Hazardous Materials at or from the Real Property owned by Seller (or, from any of the other Real Property) during Seller's occupancy of such Real Property, or arising from the operations of Seller in connection with its Real Property during Seller's occupancy of any such Real Property or its Business, in material violation of any Environmental Laws.

- (g) Seller has delivered to Buyer true and complete copies and results of any reports, studies, analyses, tests or monitoring possessed or initiated by Seller pertaining to Hazardous Materials in, on or under the Real Property, or concerning compliance by Seller, with Environmental Laws, said reports, studies, etc. to include without limitation, any and all Phase I environmental reports now or hereafter in the possession or control of Seller.
- 3.9 <u>Permits.</u> Set forth on <u>Schedule 3.9</u> is a complete and correct list of all Permits used by the Seller in the operation of the Business. To the Knowledge of the Seller, Seller has and at all relevant times has had, all Permits necessary for its operation of the Business to comply with all Laws applicable to the Assets and is and has been in material compliance with the terms of such Permits (as applicable). Seller has no Knowledge of any reason the Permits would be revoked or materially altered before the Closing Date and/or Buyer's timely request to the appropriate agency to transfer the Permit.

3.10 Insurance.

- (a) Seller maintains and has maintained appropriate insurance necessary for the full protection of all of its Assets, Business, operations, products and services. Schedule 3.10(a) sets forth a list of the insurance policies as of the date of this Agreement that insure the properties, assets or operations of the Business, or the Assets. There are no claims by Seller pending under any of such policies as to which coverage has been questioned, denied or disputed by the underwriters of such policies or in respect of which such underwriters have reserved their rights, and all such policies are in full force and effect, and no written notice of cancellation, termination or non-renewal has been received with respect to any such policy. Seller is not in default with respect to such insurance policies.
- (b) Seller has no outstanding guarantees or other credit support issued or entered into in connection with the Assets or the Business and no guarantees or other credit support is required to satisfy any contractual, statutory, or regulatory requirement applicable to Seller with respect to the Business or Assets.

3.11 Intellectual Property.

- (a) There are no registrations or applications for registration for Intellectual Property Rights owned by Seller in connection with its ownership and operation of the Business. All of the Intellectual Property Rights owned by Seller in connection with its ownership and operation of the Business are subsisting, and to the Knowledge of Seller, valid and enforceable.
- (b) To Seller's Knowledge, no Intellectual Property Right owned by Seller in connection with its ownership and operation of the Business is being infringed or misappropriated by any third party, and Seller is not infringing or misappropriating any valid and enforceable Intellectual Property Right owned by any third party, except, in any case, where such infringement or misappropriation would not result in a Material Adverse Effect.
- (c) To the Knowledge of Seller, there have been no actual or alleged theft or unauthorized disclosure, use, access, intrusions, or breaches of security, of: (i) the Systems; (ii) any personal information, payment card information, confidential or proprietary Data or any other such information collected, maintained or stored by or on behalf of Seller in connection with its ownership and operation of the Business; or (iii) any trade secrets and other confidential information that constitutes Intellectual Property Rights owned by Seller in connection with its ownership and operation of the Business, except where such theft or unauthorized disclosure, use, access, intrusions, or breaches of security would not result in a Material Adverse Effect.

- 3.12 <u>Conduct of Business in Ordinary Course</u>. Since the effective date of the Appraisal, Seller has operated the Business in the ordinary course of business consistent with past practice, and there has not been any event, occurrence or development that has had a Material Adverse Effect.
- 3.13 Proceedings. Other than as set forth on Schedule 3.13, there are no Proceedings pending or threatened against Seller or directly affecting any of the Assets or the Business by or on account of any Person or before any Governmental Authority and to Seller's Knowledge there is no valid basis for any such Proceeding. Seller has not been charged with, nor to Seller's Knowledge is it under investigation with respect to any charge which has not been resolved to their favor concerning any violation of any applicable Law with respect to any of the Assets or the Business and to Seller's Knowledge, there is no valid basis for any such charge or investigation. No judgment, Order, writ, injunction, decree, assessment or other command of any Governmental Authority affecting Seller or any of the Assets or the Business has been entered which is presently in effect, except for Orders of the IURC relating to the conduct of the Business in the ordinary course which have been disclosed to or are otherwise known to Buyer and which are included in Schedule 3.13. There is no Proceeding pending or to Seller's Knowledge, none threatened which challenges the validity of this Agreement or the Contemplated Transaction or otherwise seeks to prevent, directly or indirectly, the consummation of the Contemplated Transaction, nor to seller's Knowledge is there any valid basis for any such Proceeding.
- 3.14 Compliance with Laws. Except as set forth on Schedule 3.14, to Seller's Knowledge, it is, and has been, in compliance with all Laws applicable to the Assets and the operation of the Business and has not committed any violation of any Law applicable to the Assets and/or operation of the Business. Seller has not received any notice or other communication (whether oral or written) from any Governmental Authority or any other Person regarding (i) any actual, alleged, possible or potential violation of, or failure to comply with, any Law or (ii) any actual, alleged, possible or potential obligation on the part of Seller to undertake, or to bear all or any portion of the cost of, any remedial action of any nature. To Seller's Knowledge, the Assets, in their current condition, are capable of complying with all Laws.
- 3.15 <u>Affiliate Obligations</u>. There is no agreement, account, Contract, or transaction among Seller or any of its Affiliates that constitutes an Asset, except for customer relationships in the ordinary course of business between Seller and its directors and officers (and their respective Affiliates) who are customers of Seller.
- **3.16 Brokers.** Seller has no Liability or obligation to pay any fees or commissions to any broker, finder, or agent in connection with the negotiation, execution or delivery of this Agreement or any other Transaction Document.

ARTICLE 4 Representations and Warranties of Buyer

Buyer hereby makes the following representations and warranties to Seller as of the date of this Agreement and as of the Closing:

4.1 Organization. Buyer is a public utility corporation duly organized, validly existing and in good standing under the Laws of the State of Indiana.

4.2 Enforcement: Authority: No Conflict.

- (a) Buyer has the power to enter into and perform this Agreement and the other Transaction Documents to which it is a party and to consummate the Contemplated Transaction.
- (b) The execution, delivery, and performance of this Agreement and the other Transaction Documents to which it is a party by Buyer, and the consummation of the Contemplated Transaction, have been duly and validly authorized by all necessary action on the part of Buyer. This Agreement has been duly executed and delivered by Buyer (and all documents required to be executed and delivered by Buyer at Closing shall be duly executed and delivered by Buyer), and this Agreement constitutes, and at the Closing such documents shall constitute, the valid and binding obligations of Buyer, enforceable in accordance with their terms except as such enforceability may be limited by the Bankruptcy and Equity Exceptions.

- (c) Neither the execution, delivery and performance of this Agreement and the other Transaction Documents to which Buyer is a party by Buyer nor the consummation of the Contemplated Transaction will, directly or indirectly (with or without notice or lapse of time):
- (i) contravene, conflict with or result in a violation of: (A) any provision of the Organizational Documents of Buyer; or (B) any resolution adopted by the governing body of Buyer;
- (ii) contravene, conflict with or result in a violation of any Laws or any Order to which Buyer or any of its material assets may be subject, or require the consent of or other action by a Governmental Authority; or
- (iii) contravene, conflict with or result in a default under, or give rise to any right of termination, cancellation or acceleration of any right or obligation of Buyer or to a loss of any benefit to which Buyer is entitled under any provision of any Contract to which Buyer is a party, or by which any of the material assets and properties of Buyer are bound or subject.
- 4.3 <u>Proceedings</u>. There are no Proceedings pending or, to the Knowledge of Buyer, threatened against or affecting Buyer that challenges or seeks to prevent, enjoin, alter or materially delay the Contemplated Transaction. Buyer is not a party or subject to any Order other than orders of general applicability and is not in default thereunder.
- 4.4 <u>Brokers.</u> No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Contemplated Transaction based upon arrangements made by or on behalf of Buyer.
- 4.5 <u>Sufficiency of Funds.</u> Buyer has sufficient cash on hand or other sources of immediately available funds to enable it to make payment of the Purchase Price and consummate the Contemplated Transaction.
- 8.6 Solvency. Immediately after giving effect to the transactions contemplated hereby, Buyer shall be solvent and shall: (a) be able to pay its debts as they become due; (b) own property that has a fair saleable value greater than the amounts required to pay its debts (including a reasonable estimate of the amount of all contingent liabilities); and (c) have adequate capital to carry on its business. No transfer of property is being made and no obligation is being incurred in connection with the transactions contemplated hereby with the intent to hinder, delay or defraud either present or future creditors of Buyer or Seller. In connection with the transactions contemplated hereby, Buyer has not incurred, nor plans to incur, debts beyond its ability to pay as they become absolute and matured.
- Independent Investigation. Buyer has conducted its own independent investigation, review and analysis of the Business and the Assets, and acknowledges that it has been provided adequate access to the personnel, properties, assets, premises, books and records, and other documents and data of Seller for such purpose. Buyer acknowledges and agrees that: (a) in making its decision to enter into this Agreement and to consummate the transactions contemplated hereby, Buyer will have relied solely upon its own investigation and the express representations and warranties of Seller set forth in Article 3 of this Agreement (including related portions of the Disclosure Schedules); and (b) neither Seller nor any other Person has made any representation or warranty as to Seller, the Business, the Assets or this Agreement, except as expressly set forth in Article 3 of this Agreement (including the related portions of the Disclosure Schedules).

ARTICLE 5 Conditions Precedent to Closing

- **5.1** Conditions Precedent to the Obligations of Buyer. Buyer's obligations to consummate the Contemplated Transaction are subject to the satisfaction in full, unless expressly waived in writing by Buyer, of each of the following conditions:
- (a) Representations and Warranties. Each of the representations and warranties of Seller contained in Article 3 is true, correct and accurate from the date of this Agreement and as of the Closing Date shall be true, correct and accurate as though restated on and as of such date (except in the case of any representation and warranty that by its terms is made as of a date specified therein, which shall be accurate as of such date) except where the failure of any such representation or warranty to be true, correct and accurate would not, individually or in the aggregate, be material to the

Business; provided, however, that any representation or warranty of Seller contained in Article 3 qualified by materiality or similar qualifications shall be deemed not to be so qualified for the purposes of this Section 5.1(a);

- (b) <u>Covenants</u>. Seller shall have performed and complied in all material respects with all covenants required by this Agreement to be performed or complied with by them prior to or at the Closing;
- (c) <u>Proceedings.</u> No Order shall be in effect and no Proceeding by any Person shall be threatened or pending before any Governmental Authority, or before any arbitrator, wherein an unfavorable Order would: (i) prevent consummation of the Contemplated Transaction; (ii) have a likelihood of causing the Contemplated Transaction to be rescinded following consummation; (iii) adversely affect the right of Buyer to own any of the Assets; or (iv) adversely affect the Business prospects, value or condition of any of the Assets or the Business;
- (d) Approvals. Buyer shall have received prior to Closing (i) an Order from the IURC approving the Contemplated Transaction and the transfer of the Assets and Business thereunder, its proposed accounting and rate base treatment with respect to the Contemplated Transaction including recognition of the full Purchase Price in net original cost rate base, the application of proposed depreciation accrual rates to the Assets, the application of proposed rates, and the encumbrance of the Real Property with the lien of Buyer's mortgage indenture; and (ii) all other regulatory approvals required by any Governmental Authority to operate the Business within the Service Area;
- (e) <u>Closing Deliveries</u>. Seller shall have delivered to Buyer the Closing requirements set forth in Section 2.6(a);
- (f) <u>Due Diligence</u>. On or before 90 days after an affirmative vote by the members of Seller to approve the performance of this transaction, Buyer must have received from Seller copies of all documents Purchaser has requested pursuant to <u>Sections 6.1(b)</u> and <u>6.2(a)</u> and must have been permitted by Seller to make such investigations pursuant to <u>Section 6.1(b)</u> and <u>6.2(a)</u> as Buyer has reasonably requested, and during the thirty (30) day period after receipt of all such documents by Buyer, Buyer must not have elected by written notice to Seller based upon such documents and investigations to terminate this Agreement without consummating the Contemplated Transaction; and
- (g) No Adverse Change. Since the date of this Agreement, there has not been any event, occurrence or development which, individually or in the aggregate, has had or would have a Material Adverse Effect and there is no material adverse change in the relationships maintained by Seller with its employees, suppliers, customers or Governmental Authorities as of Closing; and
- (h) <u>Board Approval</u>. Buyer shall have obtained approval of the Contemplated Transaction by Buyer's board of directors
- **5.2** <u>Conditions Precedent to Obligations of Seller</u>. The Seller's obligation to consummate the Contemplated Transaction is subject to the satisfaction in full, unless expressly waived in writing by Seller, of each of the following conditions:
- (a) Representations and Warranties. Each of the representations and warranties of Buyer contained in Article 4 is true, correct and accurate as of the date of this Agreement and, as of the Closing Date, shall be true, correct and accurate as though restated on and as of such date (except in the case of any representation and warranty that by its terms is made as of a date specified therein, which shall be accurate as of such date) except where the failure of any such representation or warranty to be true, correct and accurate would not, individually or in the aggregate, materially impair Buyer's ability to consummate the Contemplated Transaction:
- (b) <u>Covenants</u>. Buyer shall have performed and complied in all material respects with all covenants required by this Agreement to be performed and complied with by Buyer prior to or at Closing;
- (c) <u>Closing Deliveries</u>. The Buyer shall have delivered to Seller the Closing requirements set forth in Section 2.6(b);

- (d) Approvals. Buyer shall have received prior to Closing (i) an Order from the IURC approving the Contemplated Transaction and the transfer of the Assets and Business thereunder, without material change and/or additional conditions, as determined through mutual agreement of Buyer and Seller, and (ii) all other regulatory approvals required by any Governmental Authority to operate the Business within the Service Area; and
- (e) <u>Member Approval</u>. Seller must have received prior to preparing and filing the proceeding for approval of the IURC the approval of its members to the Contemplated Transaction as required by its Organizational Documents or Indiana law.

ARTICLE 6 Covenants and Special Agreements

- **6.1** Covenants of Seller Prior to Closing. Seller covenants and agrees that during the period from the date hereof until Closing:
- (a) Non-Solicitation. Unless and until such time as this Agreement is terminated pursuant to Article 8, Seller shall not, and will cause each employee, officer, governing body, representative and agent not to, directly or indirectly: (i) submit, solicit, initiate, encourage or discuss any proposal or offer from any Person relating to any sale of all or any portion of the Assets or a sublease or assignment of any lease or any similar transaction involving Seller and the Business or the Assets; (ii) enter into any agreement or commitment related to any such transaction; or (iii) furnish any information with respect to or assist or participate in or facilitate in any other manner any effort or attempt by any Person to do or seek any of the foregoing. Seller shall notify Buyer immediately if any Person makes any proposal, offer, inquiry or contact with respect to any of the foregoing.
- (b) Access. Upon reasonable prior notice by Buyer, Seller shall: (i) furnish Buyer and its financial and legal advisors with copies of all such Contracts, books and Records and other existing documents and data as Buyer may reasonably request; (ii) furnish Buyer and its financial and legal advisors with such additional financial, operating and other data and information as Buyer may reasonably request; (iii) permit Buyer or its representatives to conduct such physical inspections and environmental audits of the Real Property, as requested by Buyer; and (iv) permit Buyer or its representatives to conduct interviews of employees of Seller.
- (c) Ordinary Course. Seller shall carry on the operation of the Business in the ordinary course of business, consistent with prior practice, not introduce any materially new method of management or operation and use reasonable efforts to preserve the Business and conserve the goodwill and relationships of Seller's customers, suppliers, Governmental Authorities and others having business relations with it. Seller shall not engage in any activity or transaction which is inconsistent with the terms of this Agreement.
- (d) <u>Liens; Encumbrances</u>. Seller shall not enter into or assume any mortgage, pledge, security agreement or other title retention agreement or permit any Encumbrance to attach to any of the Assets, whether now owned or hereafter acquired.

(e) <u>Further Covenants.</u>

- (i) Reports. Seller shall duly and timely file all reports required to be filed with any Governmental Authority and will promptly pay when due all Taxes, assessments and governmental charges including interest and penalties levied or assessed, unless diligently contested in good faith by appropriate Proceedings;
- (ii) <u>Condition of Property</u>. Consistent with past practice, Seller shall maintain and keep the Assets in substantially the same condition as of the date hereof, normal wear and tear excepted;
- (iii) <u>Insurance</u>. Seller shall maintain in full force and effect all policies of insurance now in effect up and through the Effective Time on the Closing Date;
- (iv) No Breach or Default of Contracts. Seller shall not do any act or omit any act or permit any omission to act which will cause a breach or default by Seller of any Contract;

- (v) <u>Supplies</u>. Seller shall keep supplies at a level sufficient to operate the Business in accordance with past practice;
- (vi) <u>Contracts.</u> Seller shall not enter into any Contract other than in the ordinary course of business;
- (vii) <u>Related Person Transactions</u>. Seller shall not enter into any transaction with any Related Person, except for water customer relationships in the ordinary course of business between Seller and its directors and officers and their families (and their respective Affiliates); and
- (viii) <u>Material Projects.</u> Except for repair or replacement of Assets by Seller in the ordinary course of business consistent with past practice and <u>Section 6.1(e)(ii)</u>, Seller shall provide Buyer with a timely opportunity to review and approve, or not, in its sole discretion, all material projects to be installed in the system.
- (f) <u>Updates to Disclosure Schedules</u>. Seller shall deliver to Buyer, not later than four (4) days prior to the Closing Date, written notice of supplemental information, if any, updating the information set forth in the Disclosure Schedules to reflect any occurrences or discoveries initially arising after the date of this Agreement and before the Closing Date (each, a "Disclosure Schedules Update"). Each such Disclosure Schedules Update shall (i) expressly state that it is being made pursuant to this Section 6.1(f) and (ii) describe in reasonable detail the changes, additions or events to which it relates if not otherwise readily apparent. Upon Closing, the representations and warranties of Seller shall be deemed modified and supplemented with respect to the events and matters as indicated in any Disclosure Schedules Update for all purposes of this Agreement.
- **6.2** <u>Covenants of Buyer Prior to Closing</u>. Buyer covenants and agrees that, during the period from the date hereof until Closing:
- (a) Buyer shall conduct due diligence and investigation reasonably and in good faith, and any determination or action required by Buyer as a precondition of Closing shall not be unreasonably withheld. Buyer shall promptly disclose to Seller any information discovered in Buyer's due diligence investigations that Buyer believes constitutes a breach of Seller's representations and warranties or covenants in this Agreement. If Buyer fails to disclose such information to Seller promptly after discovery by Buyer, Buyer will be deemed to have waived any claim for breach of a representation and warranty or covenant arising from such information.
- (b) Buyer shall offer employment, effective on the Closing Date, to Seller's full-time water utility employees who are employed by Seller in operating the System as of the Closing Date and who are available to commence work on the Closing Date, subject to Buyer's existing standard hiring policies and procedures applicable to new employees (including but not limited to, a criminal background check and drug screening and written acknowledgment of Buyer's Code of Conduct and other employment policies, if applicable). The personnel who accept such employment and commence employment on the Closing Date are referred to in this Agreement as the "Transferred Personnel." Buyer shall not solicit any employee of Seller prior to Closing, except as it relates to interviewing the Seller's employees for informational purposes and making the offer provided in this Section 6.2(b).
- **Environmental Assessment.** Buyer, in its sole discretion and at its sole expense, may conduct a Phase I and Phase II environmental analysis of any or all of the Real Property, and the Seller shall cooperate and provide access for same. In the event that an assessment detects any actual or possible violation of any Environmental Law, the Buyer may, in its sole discretion, terminate this Agreement the effect of which is described in <u>Section 8.2</u> herein.

6.4 Regulatory Compliance; Consents.

(a) Upon the terms and subject to the conditions set forth in this Agreement, each of Buyer and Seller, at their own respective expense, shall use its reasonable best efforts to assist, consult with and cooperate with each other and any other parties in doing all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Contemplated Transaction, including using reasonable best efforts to accomplish the following: (i) the taking of all actions necessary to cause the conditions to the Closing set forth in Article 5 to be satisfied as promptly as practicable; (ii) the obtaining of all Permits and Consents from all third parties and all

Governmental Authorities necessary or advisable to consummate, or in connection with, the Contemplated Transaction; (iii) the making of all necessary registrations and filings promptly with the appropriate Governmental Authorities, including, if applicable, any registrations and filings required by the IURC; and (iv) the execution and delivery of any additional instruments necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement.

(b) In furtherance (but not in limitation) of Section 6.4(a), Buyer and Seller shall each keep the other apprised of the status of matters relating to filings, Permits and Consents and completion of the Contemplated Transaction. Subject to applicable Law, each of Buyer and Seller shall have the right to review in advance, and, to the extent practicable, each shall consult the other on, all of the information relating to Buyer, Seller, the Business or the Assets, as the case may be, and any of their respective Affiliates, that appear in any filing made with, or written materials submitted to, any third party or any Governmental Authority in connection with the Contemplated Transaction. Buyer and Seller shall promptly (but in no event later than with respect to any required applications, notices or other filings required before the IURC or under any other applicable Law, 90 days after the date of this Agreement) make all filings and submissions with Governmental Authorities under applicable Law that are necessary or advisable to consummate, or in connection with, the Contemplated Transaction. Seller, on the one hand, and Buyer, on the other hand, shall each, in connection with the efforts referenced in this Section 6.4 to obtain all requisite Permits and Consents for the Contemplated Transaction under applicable Law, use its reasonable best efforts to: (i) cooperate in all respects with each other in connection with any filing or submission and in connection with any investigation or other inquiry, including any Proceeding initiated by a private party, in each case, regarding any such transaction; (ii) keep the other party informed of any material communication received by such party from, or given by such party to, any Governmental Authority and of any communication received or given in connection with any Proceeding by a private party, in each case regarding any such transaction; (iii) subject to applicable Law, permit the other party to review, in advance, any written communication given by it to or received from, and consult with each other in advance of any meeting or conference with, any Governmental Authority or, in connection with any Proceeding by a private party regarding any such transaction, any other Person, and to the extent permitted by any such Governmental Authority or other Person, give the other party the opportunity to attend and participate. and shall participate and attend upon the request of the other party, in such meetings and conferences subject to applicable Law; and (iv) shall, if requested by the other party or any Governmental Authority. promptly provide information or respond to questions by such Governmental Authority in connection with the subject matter of this Agreement. The parties shall each bear their own costs pursuant to this Section 6.4.

6.5 Certain Post-Closing Covenants of Seller.

(a) Seller:

- (i) shall pay in a timely manner all Taxes due from Seller resulting from or payable in connection with the sale of the Assets pursuant to this Agreement, provided such Taxes accrued prior to the Closing or as a result of the Contemplated Transaction.
- (ii) shall pay, or make adequate provisions for the payment, in full, of all of the retained Liabilities and other Liabilities of Seller under this Agreement, including the refund of customer deposits, membership fees and any proceeds from the sale of the utility assets due to members of Seller.
- (iii) hereby agrees to cooperate with Buyer to ensure a proper transition of all customers with respect to billing and customer service activities.
- (iv) hereby acknowledges that Buyer must comply with all provisions of its tariffs as filed with and approved by the IURC.
- (b) Following the Closing, Seller shall promptly deliver to Buyer any payments received from customers for water sold by Buyer to such customers following the Closing.
- (c) Subject to the obligations of Seller under applicable Law, effective as of the Closing, the Transferred Personnel shall cease active participation in the Seller's Plans. Seller shall remain liable for all eligible claims for benefits under Seller's Plans that are incurred by the Transferred Personnel prior to the Closing Date. For purposes of this Agreement, the following claims shall be deemed to be incurred as follows: (i) life, accidental death and dismemberment, short-term disability, and workers compensation insurance benefits, on the event giving rise to such benefits, (ii) medical, vision, dental, and prescription drug benefits, on the date the applicable services, materials or supplies

were provided, (iii) long-term disability benefits, on the eligibility date determined by the long-term disability insurance carrier for the plan in which the applicable Personnel participates.

(d) Subject to applicable Law and Buyer's rights and obligations set forth in Article 6, it is expressly understood that Buyer will not acquire any asset, or assume any liability or obligation in connection with the transactions contemplated by this Agreement relating to any of the Seller's Plans, Seller's Benefit Obligations, or any employment agreement. Seller shall be solely responsible for any liability, funding obligation, claim or expense arising from the Seller's Plans, Seller's Benefit Obligations, or any employment agreements, both prior to and after the Closing Date.

6.6 Certain Post-Closing Covenants of Buyer.

- (a) Buyer will, for a minimum of one year after the Closing, maintain a call center operated by the INAW Customer Support Team for customers of Seller.
- (b) Following the Closing, Buyer shall promptly deliver to Seller any payments received from customers for water sold by Seller to such customers prior to the Closing.
- (c) If, on or before the one-year anniversary of the Closing Date, Buyer terminates without cause the employment of any of the Transferred Personnel not subject to a collective bargaining agreement, or makes a materially adverse change to the compensation or other terms and conditions of employment of any Transferred Personnel not subject to a collective bargaining agreement following which such employee terminates his or her employment with Seller during such one-year period, then Buyer shall pay to such terminated employee a severance benefit equal to at least six months of such employee's compensation (at the greater of the amount established when the employee was hired pursuant to Section 6.2(b) or the amount the employee was receiving immediately prior to termination of employment). "Cause" is defined as willful or gross misconduct, moral turpitude, failure to perform duties, or breach of fiduciary duty, or any other violation of Company policies, practices, or standards. Notwithstanding Section 9.14 or any other provision of this Agreement, each Seller employee hired pursuant to Section 6.2(b) is a third-party beneficiary of the covenant in this Section 6.6(c) and is entitled to enforce it against Buyer.
- (d) Buyer shall provide each of Transferred Personnel compensation and benefits that are substantially comparable to the compensation and benefits then provided to similarly situated employees of Buyer. Nothing in this Agreement shall require Buyer to provide any particular form or type of employee benefit program, plan or policy to any Transferred Personnel as a result of the transaction contemplated by this Agreement.
- (e) Subject to the obligations of Seller under Law, Buyer's rights and obligations set forth in Article 6, and the Buyer's applicable employee benefit plan documents, with respect to employee benefit plans maintained by Buyer for the benefit of its employees (i.e., paid vacation leave, Buyer's 401k Savings Plan), effective as of the Closing, Buyer shall recognize the Transferred Personnel's length of service with the Seller as if such service were with Buyer for eligibility and vesting under Buyer's employee benefit plans and programs (except where doing so would result in a duplication of benefits). Buyer's pension plans and retiree medical plans are excluded from the foregoing sentence.
- (f) This Article 6 shall be binding upon and inure solely to the benefit of each of the Parties to this Agreement, and nothing in this Article 6, express or implied, shall confer upon any other person any rights or remedies of any nature whatsoever under or by reason of this Article 6, except for the rights provided in Section 6.6(c) and Section 6.6(e). The Parties acknowledge and agree that the terms set forth in this Article 6 shall not create any right in any Transferred Personnel or any other person to any continued employment with Buyer or compensation or benefits of any nature or kind whatsoever, and shall not be deemed to restrict Buyer in the exercise of its independent business judgment in establishing or modifying any terms or conditions of the employment of the Transferred Personnel. Nothing contained in this Article 6 shall constitute an amendment of, or an undertaking to amend, any employee benefit plan, program or arrangement maintained by Buyer or is intended to prevent Buyer from amending or terminating any such employee benefit plan, program or arrangement in accordance with its terms.

ARTICLE 7 Indemnification; Limitations

- **7.1 Survival.** Subject to the limitations and other provisions of this Agreement, the representations, warranties, covenants and obligations contained herein shall survive the Closing and shall remain in full force and effect until the last day of the calendar year that includes the Closing Date; provided, that the representations and warranties in Section 3.1, Section 3.2, Section 3.3(a), and Section 3.4 shall survive indefinitely. Notwithstanding the foregoing, any claims asserted in good faith with reasonable specificity (to the extent known at such time) and in writing by notice from the non-breaching party to the breaching party prior to the expiration date of the applicable survival period shall not thereafter be barred by the expiration of the relevant representation or warranty and such claims shall survive until finally resolved.
- Indemnification and Payment of Damages by Seller. Subject to the other terms and conditions of this Article and Section 6.2(a), Seller shall defend, indemnify, and hold harmless Buyer and Buyer's past, present and future officers, directors, shareholders, employees, agents, attorneys, representatives, successors and assigns (collectively, the "Buyer Indemnitees"), from any and all Damages arising out of, or caused by: (i) the misrepresentation, breach of warranty or nonfulfillment of any provision of this Agreement by Seller or; (ii) all Liabilities and/or duties of Seller, whether accruing prior to or after the Effective Time on the Closing Date, and any Encumbrance affecting the Assets; (iii) assessments, charges and other similar claims due or owing, directly or indirectly, by Seller or otherwise as a result of or on account of the Assets or the Business at any time prior to the Effective Time on the Closing Date; (iv) the ownership and/or operation of any of the Assets or the Business prior to the Effective Time on the Closing Date; (v) any claim or Proceeding now existing or hereafter arising and relating to the Assets or the Business of Seller and arising from events or matters occurring prior to the Effective Time on the Closing Date; and (vi) any claim by an employee of Seller for any severance payment or arising out of such employee's employment with Seller or under the Worker Adjustment and Retraining Notification Act, COBRA (Sections 601 through 608 of the Employee Retirement Income Security Act of 1974), or under any employee benefit plan or employment Contract to which Seller is a party.
- 7.3 Indemnification and Payment of Damages by Seller Environmental Matters. In addition to the provisions of Section 7.2, subject to the other terms and conditions of this Article and Section 6.2(a), Seller shall defend, indemnify, and hold harmless Buyer Indemnitees, from any and all Damages (including costs of cleanup, containment or other remediation) arising out of, or caused by:
- (a) Any Liabilities under Environmental Law arising out of or relating to: (i) (A) the ownership, operation or condition at any time on or prior to the Effective Time on the Closing Date of the Assets or any other properties (whether real, personal or mixed and whether tangible or intangible) in which Seller has or had an interest; or (B) any Hazardous Materials or other contaminants that were present on or in the Assets at any time on or prior to the Effective Time on the Closing Date; or (ii) any Hazardous Materials or other contaminants, wherever located, that were, or were allegedly, generated, transported, stored, treated, Released or otherwise handled by Seller or by any other Person for whose conduct Seller is or may be held responsible, at any time on or prior to the Effective Time on the Closing Date; or
- (b) Any bodily injury (including illness, disability and death), personal injury, property damage (including trespass, nuisance, wrongful eviction and deprivation of the use of real property) or other damage of or to any Person, including any employee or former employee of Seller or any other Person for whose conduct it is or may be held responsible, in any way arising from or allegedly arising from Hazardous Material that was: (i) present or suspected to be present on or before the Effective Time on the Closing Date on or at the Real Property (or present or suspected to be present on any other property, if such Hazardous Material emanated or allegedly emanated from any of the Real Property and was present or suspected to be present on any of the Real Property on or prior to the Effective Time on the Closing Date); or (ii) Released or allegedly Released by Seller or any other Person for whose conduct it is or may be held responsible, at any time on or prior to the Effective Time on the Closing Date.

Buyer shall control any cleanup, any related Proceeding and, except as provided in the following sentence, any other Proceeding with respect to which indemnity may be sought under this Section 7.3. The procedure described in Section 7.5 will apply to any claim solely for monetary Damages relating to a matter covered by this Section 7.3.

7.4 Indemnification By Buyer. Subject to the other terms and conditions of this Article, Buyer shall defend, indemnify, and hold harmless Seller and Seller's past, present and future officers, governing body, employees, agents, attorneys, representatives, successors and assigns from any and all Damages arising out of, or caused by: (i) Buyer's misrepresentation, breach of warranty or nonfulfillment of any provision of this Agreement; (ii) any claim or Proceeding arising after the Effective Time on the Closing Date and relating to events or matters occurring subsequent to the Effective Time on the Closing Date; and (iv) any claim by an employee of Buyer arising out of such employee's employment with Buyer after the Effective Time on the Closing Date.

7.5 Procedure for Indemnification. The procedure for indemnification shall be as follows:

- (a) The party claiming indemnification (the "Claimant") shall promptly give notice to the party from whom indemnification is claimed (the "Indemnifying Party") of any claim, whether between the parties or brought by a third party, specifying: (i) the factual basis for such claim; and (ii) the amount of the claim (to the extent known). If the claim relates to a Proceeding filed by a third party against Claimant, Claimant shall give such notice within ten (10) Business Days after written notice of such Proceeding was given to Claimant. Claimant's failure to give the Indemnifying Party such notice shall not preclude Claimant from obtaining indemnification from the Indemnifying Party unless Claimant's failure has materially prejudiced the Indemnifying Party's ability to defend the claim or litigation, and then the Indemnifying Party's obligation shall be reduced to the extent of such prejudice.
- (b) Following receipt of notice from the Claimant of a claim between the parties, the Indemnifying Party shall have sixty (60) days to make such investigation of the claim as the Indemnifying Party deems necessary or desirable. For the purposes of such investigation, the Claimant agrees to make available to the Indemnifying Party and/or its authorized representatives the information relied upon by the Claimant to substantiate the claim. If the Claimant and the Indemnifying Party agree at or prior to the expiration of said sixty (60)-day period (or any mutually agreed upon extension thereof) to the validity and amount of such claim, the Indemnifying Party shall immediately pay to the Claimant the full amount of the claim. If the Claimant and the Indemnifying Party do not agree within said period (or any mutually agreed upon extension thereof), the Claimant may seek appropriate legal remedy.
- (c) With respect to any claim by a third party as to which the Claimant asserts it is entitled to indemnification hereunder, the Indemnifying Party shall have the right, at its own expense, to participate in or at its election to assume control of the defense of such claim, with counsel reasonably satisfactory to Claimant, subject to reimbursement of Claimant for actual out-of-pocket expenses incurred by Claimant as the result of request by the Indemnifying Party, subject to the following:
- (i) The Claimant may retain separate co-counsel at its sole cost and expense and participate in the defense of any such claim by a third party; and
- (ii) The Indemnifying Party shall conduct the defense of the third-party claim actively and diligently thereafter.

If the Indemnifying Party elects to assume control of the defense of any third party claim pursuant to this Section, the Indemnifying Party may nevertheless reserve the right to dispute the amount of indemnification claimed or dispute Claimant's right to be indemnified with respect to all or any portion of the claim. Except with the written Consent of the Claimant, the Indemnifying Party shall not, in defending any claim or any litigation resulting therefrom, consent to entry of any judgment or enter into any settlement which does not release the Claimant from all Liability in respect of such claim or litigation. In the event the Claimant fails to consent to any settlement or compromise which such failure results in Damages in excess of the amount for which Consent was requested, the limitation of the Indemnifying Party's obligations to indemnify the Claimant with respect to the subject matter of the claim shall be the amount of the proposed settlement or compromise rejected by Claimant and the Claimant shall be responsible for, and shall hold harmless the Indemnifying Party from, all Damages (including, without limitation, reasonable attorneys' fees incurred with respect to matters subsequent to the rejection of the settlement by Claimant) in excess of the amount of the proposed settlement or compromise rejected by Claimant.

(d) If a claim, whether between the parties or by a third party, requires immediate action, the parties will make every effort to reach a decision with respect thereto as expeditiously as possible.

7.6 <u>Limitations on Damages</u>.

- (a) The Indemnifying Party shall not be liable to any Claimant for indemnification under Section 7.2 or Section 7.3, as the case may be, until the aggregate amount of all Damages indemnifiable under Section 7.2 or Section 7.3 exceeds one million dollars (\$1,000,000) (the "Deductible"), in which event the Indemnifying Party shall only be required to pay or be liable for Losses in excess of the Deductible. Notwithstanding the preceding sentence, the Deductible does not apply to any claim under Section 6.6(c).
- (b) The aggregate amount of all Damages for which Seller shall be liable pursuant to <u>Section 7.2</u> and <u>Section 7.3</u> shall not exceed thirty percent (30%) of the Purchase Price, except to the extent the Damages arise from Fraud committed by Seller.
- (c) Except to the extent such damages arise from Fraud, in no other event shall any Indemnifying Party be liable to any Claimant for any incidental, consequential, special or indirect damages, including loss of future revenue or income, loss of business reputation or opportunity relating to the breach or alleged breach of this Agreement, or diminution of value, nor shall any Indemnifying Party be liable to any Claimant for any punitive damages or any damages based on any type of multiple, except to the extent actually awarded to a Governmental Authority or other third party.
- (d) Each Claimant party shall take, and cause its Affiliates to take, all reasonable steps to mitigate any Damages upon becoming aware of any event or circumstance that would be reasonably expected to, or does, give rise thereto, including incurring costs only to the minimum extent necessary to remedy the breach that gives rise to such Damages. In no event shall the Indemnifying Party be liable to any Claimant for any Damages incurred as a result of the Claimant's failure to mitigate Damages.
- (e) Seller shall not be liable under this Article for any Damages based upon or arising out of any inaccuracy in or breach of any of the representations or warranties of Seller contained in this Agreement if Buyer had actual Knowledge of such inaccuracy or breach prior to the Closing.

ARTICLE 8 Termination

- **8.1** Termination and Abandonment. This Agreement may be terminated and abandoned at any time prior to the Closing Date:
 - (a) by mutual written Consent of Buyer and Seller; or
- (b) by Buyer or Seller, if the IURC does not approve the Contemplated Transaction by one year following the filing date for approval or such other later date as agreed upon, in writing, by the parties; or
- (c) by Buyer or Seller, if Seller's members do not approve the Contemplated Transaction by three months following the date Seller first requests such approval or such other later date as agreed upon, in writing, by the parties.
- **8.2 Effect of Termination.** The right of each party to terminate this Agreement under Section 8.1 is in addition to any other rights such party may have under this Agreement or otherwise, and the exercise of a right of termination will not be an election of remedies. If this Agreement is terminated pursuant to Section 8.1, all further obligations of the parties under this Agreement will terminate, except that the obligations set forth in Sections 9.9 (Legal Fees, Costs) and 9.15 (Publicity; Announcements); and all other covenants and agreements which by their terms continue after the termination of this Agreement will survive; provided, however, that if this Agreement is terminated by a party because of the breach of the Agreement by another party or because one (1) or more of the conditions to the terminating party's obligations under this Agreement, the terminating party's right to pursue all legal remedies will survive such termination unimpaired.

ARTICLE 9 General Provisions

- **9.1** Amendment and Modification. No amendment, modification, supplement, or termination of any section or provision of this Agreement will in any event be effective unless the same is in writing and is signed by the parties. Any Consent given pursuant this Agreement, any waiver of any provision of this Agreement, or any Consent to any departure from the terms of any provision of this Agreement shall be effective only in the specific instance and for the specific purpose for which given and only if it is given in writing and is signed by the party granting such Consent or waiver.
- **9.2** Assignments. Neither party may assign or transfer any of its rights or obligations under this Agreement to any other Person without the prior written Consent of the other party.
- **9.3** Captions. Captions contained in this Agreement and any table of contents preceding this Agreement have been inserted herein only as a matter of convenience and in no way define, limit, extend or describe the scope of this Agreement or the intent of any provision hereof.
- 9.4 Counterparts: Electronic Mail. This Agreement may be executed by the parties hereto on any number of separate counterparts, and all such counterparts so executed constitute one agreement binding on all the parties hereto notwithstanding that all the parties hereto are not signatories to the same counterpart. For purposes of this Agreement, a document (or signature page thereto) signed and transmitted in .pdf format by electronic mail is to be treated as an original document. The signature of any party thereon is to be considered as an original signature, and the document transmitted is to be considered to have the same binding effect as an original signature on an original document. At the request of any party hereto, the .pdf copy is to be re-executed in original form by the parties who executed the .pdf copy. No party hereto may raise the use of a .pdf copy or the fact that any signature was transmitted through the use of electronic mail as a defense to the enforcement of this Agreement or any amendment or other document executed in compliance with this Section 9.4.
- **9.5** Entire Agreement. This Agreement and the other Transaction Documents constitute the entire agreement among the parties hereto pertaining to the subject matter hereof and supersede all prior agreements, letters of intent, understandings, negotiations and discussions of the parties hereto, whether oral or written.
- **9.6** Exhibits and Schedules. All of the Exhibits and Schedules attached to this Agreement or contained in the Disclosure Schedules are deemed incorporated herein by reference.
- **9.7** Failure or Delay. Except as otherwise provided by this Agreement, no failure on the part of any party hereto to exercise, and no delay in exercising, any right, power or privilege hereunder operate as a waiver thereof; nor does any single or partial exercise of any right, power or privilege hereunder preclude any other or further exercise thereof, or the exercise of any other right, power or privilege. No notice to or demand on any party hereto in any case entitles such party to any other or further notice or demand in similar or other circumstances.
- 9.8 Governing Law. This Agreement and the rights and obligations of the parties hereunder are to be governed by and construed and interpreted in accordance with the Laws of the State of Indiana applicable to Contracts made and to be performed wholly within Indiana, without regard to choice or conflict of Laws rules. In the event of any litigation or claim regarding this Agreement, the parties agree that the IURC has jurisdiction to govern all matters involving the Contemplated Transaction and the provision of water service by Buyer to the residents located within the Service Area.
- **9.9** Legal Fees. Costs. All legal, consulting and advisory fees and other costs and expenses incurred in connection with this Agreement and the Contemplated Transaction are to be paid by the party incurring such costs and expenses.
- **9.10** Notices. All notices, Consents, requests, demands and other communications hereunder are to be in writing and are deemed to have been duly given, made or delivered: (i) when delivered in person, (ii) three (3) days after deposited in the United States mail, first class postage prepaid, (iii) in the case of telegraph or overnight courier services, one (1) Business Day after delivery to the telegraph company or overnight courier service with payment provided, or (iv) in the case of electronic mail, when sent, verification received, in each case addressed as follows:

if to Seller:

Attn: President Silver Creek Water Corporation 8104 County Line Rd. Sellersburg, IN 47172

with a copy to (which shall not constitute notice):
J. Christopher Janak
Bose McKinney& Evans LLP
111 Monument Circle, Suite 2700
Indianapolis, IN 46204

if to Buyer:

Attn: President Indiana-American Water Company, Inc. 153 N. Emerson Avenue Greenwood, Indiana 46143

with a copy to (which shall not constitute notice):
Attn: Corporate Counsel
Indiana-American Water Company, Inc.
153 N. Emerson Avenue
Greenwood, Indiana 46143

or to such other address as any party hereto may designate by notice to the other parties in accordance with the terms of this Section 9.10. For e-mail, a party shall contact the other party to receive the then-applicable e-mail for the person in the position stated above.

- 9.11 Severability. Any provision of this Agreement which is prohibited, unenforceable or not authorized in any jurisdiction is, as to such jurisdiction, ineffective to the extent of any such prohibition, unenforceability or non-authorization without invalidating the remaining provisions hereof, or affecting the validity, enforceability or legality of such provision in any other jurisdiction, unless the ineffectiveness of such provision would result in such a material change as to cause completion of the Contemplated Transaction to be unreasonable.
- 9.12 Specific Performance and Injunctive Relief. The parties hereto recognize that if any or all of them fail to perform, observe or discharge any of their respective obligations under this Agreement, a remedy at Law may not provide adequate relief to the other parties hereto. Therefore, in addition to any other remedy provided for in this Agreement or under applicable Law, any party hereto may demand specific performance of this Agreement, and such party shall be entitled to temporary and permanent injunctive relief, in a court of competent jurisdiction at any time when any of the other parties hereto fail to comply with any of the provisions of this Agreement applicable to such party. To the extent permitted by applicable Law, all parties hereto hereby irrevocably waive any defense based on the adequacy of a remedy at Law which might be asserted as a bar to such party's remedy of specific performance or injunctive relief.
- **9.13** Successors and Assigns. Subject to Section 9.2, all provisions of this Agreement are binding upon, inure to the benefit of and are enforceable by or against the parties hereto and their respective heirs, executors, administrators or other legal representatives and permitted successors and assigns.
- **9.14** No Third-Party Beneficiary. This Agreement is solely for the benefit of the parties hereto and their respective successors and permitted assigns, and no other Person has any right, benefit, priority or interest under, or because of the existence of, this Agreement, except for Persons specifically entitled to indemnification under this Agreement as identified in Section 6.6
- **9.15** Publicity: Announcements. From the date hereof through and including Closing, no party hereto shall issue, cause or permit the publication of, by any of their respective Related Persons, agents or representatives, any press release or other public announcement with respect to this Agreement or the Contemplated Transaction except: (i) with the Consent of the other parties hereto (which shall not be unreasonably withheld); (ii) as required by applicable Law (including, without limitation, any applicable securities Law); or (iii) with respect to Seller, as reasonably required for Seller

to solicit the approval of the Contemplated Transaction by Seller's members. Seller will not, without the prior Consent of Buyer, make any announcements to employees of Seller with respect to the Contemplated Transaction and, at such time as an announcement to the employees is made, Buyer shall be allowed to participate in such announcement.

9.16 <u>Cooperation.</u> The parties shall use good faith in the administration and execution of this Agreement. Any notices or certifications given under this Agreement or any related agreement shall be given in good faith without any intention to unfairly impede or delay the other party. Buyer and Seller shall cooperate fully with each other and their respective counsel and accountants in connection with any actions required to be taken as part of their respective obligations under this Agreement including, without limitation, actions required to be taken with respect to obtaining any applicable regulatory approval of the Contemplated Transaction. Buyer and Seller shall execute such other documents as may be necessary and desirable to the implementation and consummation of this Agreement. Each party agrees to use all reasonable efforts to consummate the Contemplated Transaction including, without limitation, doing all things reasonably necessary to obtain the requisite regulatory approval.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first above written.

Buyer	Seller
INDIANA-AMERICAN WATER COMPANY, INC., an Indiana public utility corporation	SILVER CREEK WATER CORPORATION an Indiana nonprofit corporation
By: Mark	By: Day limb
Name: <u>Matthew Prine</u> Title: <u>President</u>	Name: Doug Chumbler Title: President

4571436.5

Silver Creek Water Corporation
System Map Valves = 2,380 Hydrants = 952 Item Summary 816,431 ft 154,6mi Piping Summary 1" = 1,000'

Service Area

Exhibit B

Definitions

"Affiliate" means, with respect to any Person, a Person that directly or indirectly controls, is controlled by, or is under common control with, such Person, with control in such context meaning the ability to direct the management or policies of a Person through ownership of voting shares or other securities, pursuant to a written agreement, or otherwise.

"Assets" as defined in Section 2.1.

"Agreement" as defined in the introductory paragraph.

"Assigned Contracts" as defined in Section 2.1(d).

"Assignment and Assumption Agreement" as defined in Section 2.6(a)(vii).

"Bankruptcy and Equity Exceptions" as defined in Section 3.2(b).

"Bill of Sale" as defined in Section 2.6(a)(i).

"Business" as defined in the Recitals.

"Business Days" means any day other than (i) Saturday or Sunday, or (ii) any other day on which governmental offices in the State of Indiana are permitted or required to be closed.

"Buyer" as defined in the introductory paragraph.

"Closing" as defined in Section 2.5.

"Closing Date" as defined in Section 2.5.

"Consent" means any approval, consent, ratification, waiver or other authorization.

"Contemplated Transaction" means all of the transactions contemplated by this Agreement and the Transaction Documents.

"Contract" means any written or oral binding contract, commitment, lease, license, mortgage, bond, note or other instrument, or other legally binding agreement, and all amendments thereto, but excluding any Permits.

<u>"Customer Premises"</u> means a dwelling, building, structure or parcel of real estate which is supplied with water service through a Service Line.

<u>"Customer Service Connection"</u> means that portion of water pipe extending from the Customer Premises to the curb box containing the curb service stop which Customer Service Connection shall be owned and maintained by the customer.

"<u>Damages</u>" means any actions, suits, debts, sums of money, interest owed, accounts, controversies, agreements, charges, damages, judgments, executions, and reasonably incurred costs, expenses, fees (including reasonable attorneys' fees and court costs), counterclaims, claims, demands, causes of action, liabilities and losses; *provided, however*, that "Damages" shall not include any incidental, consequential, special or indirect damages, including loss of future revenue or income, loss of business reputation or opportunity relating to the breach or alleged breach of this Agreement, or diminution of value, nor shall it include punitive damages or any damages based on any type of multiple, except to the extent actually awarded to a Governmental Authority or other third party.

"<u>Data</u>" means the data relating to the Business as currently stored in an electronic format on computer servers operated by Seller, including financial, customer payment and billing information, customer service records, and maintenance records.

"Effective Time" as defined in Section 2.5.

"Encumbrance" means, with respect to any property or asset, any charge, claim, community property interest, condition, easement, encumbrance, equitable interest, lien, mortgage, option, pledge, security interest, right of first refusal, right of way, servitude or restriction of any kind in respect of such property or asset, including any restriction on use, transfer, receipt of income or exercise of any other attribute of ownership, or any repayment obligation under any grant.

"Environment" means soil, land surface or subsurface strata, surface waters (including navigable waters, ocean waters, streams, ponds, drainage basins and wetlands), groundwater, drinking water supply, stream sediments, ambient air (including indoor air), plant and animal life and any other environmental medium or natural resource.

"Environmental Law" means any Federal, State, County or Municipal laws, statutes and implementing regulations relating to environmental protection, health, safety or natural resources, including without limitation CERCLA, RCRA, the Emergency Planning and Community Right-to-Know Act (42 U.S.C. 11001 et seq.), the Hazardous Materials Transportation Act (49 U.S.C. 1801 et seq.), the Clean Water Act (33 U.S.C. 1251 et seq.), the Safe Drinking Water Act (42 U.S.C. §300f et seq.), and the Clean Air Act (42 U.S.C. 7401 et seq.).

"Excluded Assets" as defined in Section 2.2.

"Fraud" means an act, committed by a party, with intent to deceive another party and requires (a) a false representation of material fact made in Article 4 by such party; (b) with actual knowledge that such representation is false; (c) with an intention to induce the party to whom such representation is made to act or refrain from acting in reliance upon it; (d) causing that party, in justifiable reliance upon such false representation and with ignorance to the falsity of such representation, to take or refrain from taking action; and (e) causing such party to suffer material loss by reason of such reliance; provided, however, that for the avoidance of doubt, "Fraud" shall not include any type of constructive or equitable fraud.

"Governmental Authority(ies)" means any federal, state or local government located in the United States, including any political subdivision, departments, courts, commissions, boards, bureaus, ministries, agencies, or other instrumentalities of any of them.

"Hazardous Materials" means any pollutant, toxic substance, including asbestos and asbestoscontaining materials, hazardous waste, hazardous material, hazardous substance, contaminant, petroleum and petroleum-containing materials, radiation and radioactive materials, leaded paints, toxic mold or other harmful biological agents, and polychlorinated biphenyls as defined in, the subject of, or that could give rise to Liability under any Environmental Law.

"IDEM" as defined in Section 3.2(c)(ii).

"Intellectual Property Rights" means all of the following in all jurisdictions throughout the world: (i) patents, patent applications, patent or invention disclosures, design rights, and inventions (whether or not patentable and whether or not reduced to practice) and any reissue, continuation, continuation in part, division, extension or reexamination thereof; (ii) trademarks, trade names, service marks, trade dress, corporate names slogans and other indicia of source or origin, together with all goodwill associated therewith, and all translations, adaptations, derivations and combinations of the foregoing (and all logos related to the foregoing), and all registrations and applications therefor; (iii) copyrights, copyrightable or copyrighted works and mask work, and all registrations and applications therefor; (iv) internet domain names and social media accounts; (v) trade secrets and other confidential information, ideas, know how, related processes and techniques, manufacturing processes, research and development information,

drawings, specifications, formulas, designs, plans, proposals, technical data and manuals, internal business information, identities of, individual requirements of, specific contractual arrangements with, and information about, suppliers, distributors, customers, independent contractors or other business relations, compilations of data and analyses, systems, records, reports, documentation, models, innovations, improvements, methods, designs, analyses, reports and all similar or related information; (vi) computer programs and software (including source code, object code, executable code, firmware, systems, tools, Data, databases, and related documentation) and software implementations of algorithms, models and methodologies; (vii) databases, (viii) moral rights and (ix) all other intellectual property and intangible properties.

"IURC" as defined in Section 2.5.

"Knowledge" means an individual will be deemed to have "Knowledge" of a particular fact or other matter if:

- (a) Such individual is actually aware of such fact or other matter; or
- (b) A prudent individual could be expected to discover or otherwise become aware of such fact or other matter in the course of conducting a reasonable investigation concerning the existence of such fact or other matter.

A Person (other than an individual) will be deemed to have "Knowledge" of a particular fact or other matter if any individual who is serving as a director or officer of such Person (or in any similar executive capacity) has, or at any time had, Knowledge of such fact or other matter.

"<u>Law</u>" means any law, statute, rule, regulation, ordinance, order, decree, requirement, judgment, and code of any Governmental Authority.

"Lease" as defined in Section 3.4(b).

"Leased Real Property" as defined in Section 3.4(b).

"<u>Liability</u>" means, with respect to any Person, any liability or obligation of such Person, whether known or unknown, absolute or contingent, accrued or unaccrued, disputed or undisputed, liquidated or unliquidated, secured or unsecured, joint or several, due or to become due, vested or unvested, executory, determined, determinable or otherwise and whether or not the same is required to be accrued on the financial statements of such Person.

"Material Adverse Effect" means any change, condition or event that is materially adverse to (a) the operations, results of operations or financial condition of the Assets or the Business, taken as a whole or (b) the ability of Seller to consummate the transactions contemplated hereby; provided, however, that "Material Adverse Effect" shall not include any event, occurrence, fact, condition or change, directly or indirectly, arising out of or attributable to: (i) general economic or political conditions; (ii) conditions generally affecting the industries in which the Business operates; (iii) any changes in financial, banking or securities markets in general, including any disruption thereof and any decline in the price of any security or any market index or any change in prevailing interest rates; (iv) acts of war (whether or not declared), armed hostilities or terrorism, or the escalation or worsening thereof; (v) any action required or permitted by this Agreement or any action taken (or omitted to be taken) with the written consent of or at the written request of Buyer; (vi) any matter of which Buyer is aware on the date hereof; (vii) any changes in applicable Laws or accounting rules (including GAAP) or the enforcement, implementation or interpretation thereof; (viii) the announcement, pendency or completion of the transactions contemplated by this Agreement, including losses or threatened losses of employees, customers, suppliers, distributors or others having relationships with the Seller and the Business; (ix) any natural or man-made disaster or acts of God; (x) any epidemics, pandemics, disease outbreaks, or other public health emergencies; or (xi) any failure by the Business to meet any internal or published projections, forecasts or revenue or earnings predictions.

"Material Contracts" as defined in Section 3.7.

"Order" means any award, decision, injunction, judgment, order, ruling, subpoena or verdict entered, issued, made or rendered by any Governmental Authority or by any arbitrator.

"Organizational Documents" means the articles or certificate of incorporation, the bylaws or other similar organizational or governing documents of a corporation and any amendment thereto.

"Owned Real Property" as defined in Section 3.4(a).

"Permits" means all permits, licenses, authorizations, consents and approvals from or of, and has made all material filings, applications and registrations with, all Governmental Authorities required to own and operate the Business as now conducted.

"Permitted Encumbrance" means:

- (a) mechanic's, materialman's, carrier's, repairer's and other similar Encumbrances arising or incurred in the ordinary course of business or that are not yet due and payable or are being contested in good faith and by appropriate proceedings and for which reserves have been established on the Balance Sheet;
- (b) Encumbrances in the nature of zoning restrictions, easements, rights or restrictions of record on the use of real property that do not materially impair the continued use of such property in the Business in the manner in which it is currently used;
- (c) Encumbrances to secure obligations owed to landlords, lessors or renters under leases or rental agreements for the occupancy or use of real or personal property;
- (d) Encumbrances that will be fully and unconditionally discharged on or prior to the Closing Date:
 - (e) Intellectual Property Rights licenses; and
 - (f) other non-monetary and non-financial Encumbrances which would not, individually or in the aggregate, materially impair the continued use of such property and asset in the manner in which it is currently used.

"Person" means any individual, corporation (including any non-profit corporation), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, labor union or other entity or Governmental Authority.

"Proceeding" means any action, arbitration, hearing, litigation, suit, claim or other similar proceeding commenced, brought, conducted or heard by or before, or otherwise involving, any Governmental Authority or arbitrator.

"Purchase Price" as defined in Section 2.3.

"Real Property" as defined in Section 3.4(a) and (b).

"<u>Records</u>" means information that is inscribed on a tangible medium or that is stored in an electronic or other medium and is retrievable in perceivable form.

"Related Person" with respect to a particular individual, means:

- (a) Each other member of such individual's Family (as hereinafter defined);
- (b) Any Person that is directly or indirectly controlled by such individual or one or

more members of such individual's Family;

- (c) Any Person in which such individual or members of such individual's Family hold (individually or in the aggregate) a Material Interest; and
- (d) Any Person with respect to which such individual or one or more members of such individual's Family serves as a director, officer, partner, executor or trustee (or in a similar capacity).

With respect to a specified Person other than an individual:

- (e) Any Person that directly or indirectly controls, is directly or indirectly controlled by, or is directly or indirectly under common control with such specified Person;
 - (f) Any Person that holds a Material Interest in such specified Person;
- (g) Each Person that serves as a director, officer, partner, executor or trustee of such specified Person (or in a similar capacity);
 - (h) Any Person in which such specified Person holds a Material Interest;
- (i) Any Person with respect to which such specified Person serves as a general partner or a trustee (or in a similar capacity); and
 - (j) Any Related Person of any individual described in clause (b) or (c).

For purposes of this definition, (i) the "Family" of an individual includes (A) the individual, (B) the individual's spouse, (C) any other natural person who is related to the individual or the individual's spouse within the second degree, and (D) any other natural person who resides with such individual; and "Material Interest" means direct or indirect beneficial ownership (as defined in Rule 13d-3 under the Securities Exchange Act of 1934) of voting securities or other voting interests representing at least five percent (5%) of the outstanding equity securities or equity interests in a Person.

"Release" means any releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, migrating, disposing or dumping of Hazardous Materials into the environment (including, without limitation, the abandonment or discarding of barrels, containers and other closed receptacles containing any Hazardous Materials) and any condition that results in the exposure of a person to a Hazardous Materials.

"Seller" shall have the meaning given that term in the introductory paragraph.

"Seller's Benefit Obligations" means all material obligations, arrangements, or practices, whether or not reduced to writing, funded or legally enforceable, to provide benefits, other than salary or wages to present or former directors, employees or agents, (other than obligations, arrangements and practices that are Seller's Plans), that are owed, adopted or followed by the Seller. Seller's Benefit Obligations also include consulting agreements under which the compensation paid does not depend upon the amount of service rendered, sabbatical policies, severance payment policies and fringe benefits within the meaning of Code §132, which are not otherwise a Seller's Plan.

"Seller's Plans" means each voluntary employees' beneficiary association under Section 501(c)(9) of the Code whose members include any Personnel and any employee benefit plans or any other retirement, pension, profit sharing, stock option, other post-employment benefits (OPEB) stock bonus, deferred compensation (including any "nonqualified deferred compensation plan" within the meaning of Section 409A of the Code), severance, sick leave or other material plan or arrangement, in each case whether or not reduced to writing and whether funded or unfunded, including each "employee benefit plan" within the meaning of Section 3(3) of ERISA, whether or not tax-qualified and whether or not subject to ERISA, which is or has been maintained, sponsored, contributed to, or required to be contributed to by Seller for the benefit of any current or former employee, officer, director, manager, retiree, independent

contractor or consultant of the Seller or any spouse or dependent of such individual, or under which Seller has or may have any Liability.

"Service Area" as defined in the Recitals.

<u>"Service Line"</u> means that portion of water pipe extending from the water main to and including the curb stop and curb box located at or near the property line of a Customer Premises.

"Systems" means the computer systems (including the computer software, firmware and hardware), telecommunications, networks, peripherals, platforms and other similar or related items of automated, computerized and/or software systems that are used or relied on by Seller in connection with its ownership and operation of the Business.

"Tangible Personal Property" means all collection, pumping, and treatment systems, equipment, tools, furniture, office equipment, computer hardware, supplies (including chemicals and spare parts), materials, vehicles and other items of tangible personal property of every kind owned or leased by Seller (wherever located and whether or not carried on Seller's books), which are, could be, or in the future would be utilized to provide water service to Seller's water customers, together with any express or implied warranty by the manufacturers or sellers or lessors of any item or component part thereof, and all maintenance Records and other documents relating thereto.

"Tax" means (i) federal, state, or local taxes, charges, fees imposts, levies, or other assessments, including all net income, gross receipts, franchise, capital, sales, use, ad valorem, value added, transfer, profits, inventory, capital stock, license, withholding, payroll, employment, social security, unemployment, excise, severance, stamp, occupation, property and estimated taxes, fees, assessments, and charges of any kind whatsoever; (ii) all interest, penalties, fines, additions to tax, or additional amounts imposed by any Governmental Authority in connection with any item described in subsection (i); and (iii) any Liability for any item described in subsections (i) and (ii), payable by reason of Contract, assumption, transferee Liability, operation of Law, or otherwise.

"<u>Transaction Documents</u>" means this Agreement, the Bill of Sale, the Assignment and Assumption Agreement and all other documents, certificates, assignments, instruments and agreements related to this Agreement or executed and delivered in connection with this Agreement, as the same may be amended, restated, modified or otherwise replaced by mutual agreement from time to time.

Rules of Construction

For purposes of this Agreement and the other documents executed in connection herewith, the following rules of construction shall apply, unless specifically indicated to the contrary: (i) wherever from the context it appears appropriate, each term stated in either the singular or plural shall include the singular and the plural, and pronouns stated in the masculine, feminine or neuter gender shall include the masculine, the feminine and the neuter; (ii) the term "or" is not exclusive; (iii) the term "including" (or any form thereof) shall not be limiting or exclusive; (iv) all references to statutes and related regulations shall include any amendments of same and any successor statutes and regulations; (v) all references in this Agreement or in the Schedules to this Agreement to sections, schedules, exhibits and attachments shall refer to the corresponding sections, schedules, exhibits and attachments of or to this Agreement; and (vi) all references to any instruments or agreements, including references to any of the documents executed in connection herewith, shall include any and all modifications or amendments thereto and any and all extensions or renewals thereof.

This Agreement is the product of a negotiation. The parties each had an equal role in drafting the Agreement. As a result, in the event of any interpretation or construction, no term or provision shall be construed against or in favor of either of the parties on the basis of one party being deemed the drafter or based on authorship.